

Format for ANSWERING REVIEWERS



Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 10376-review.doc).

Title: Preimplantation HLA typing: Practical tool for stem cell transplantation treatment of congenital disorders

Author: Anver Kuliev, Svetlana Rechitsky

Name of Journal: *World Journal of Medical Genetics*

ESPS Manuscript NO: 10376

The manuscript has been improved according to the suggestions of reviewers:

As there were no comments from both of reviewers, I have made the update and revisions, highlighted in red, according to the Journal requirements.

Please also find attached the signed Copyright Assignment.

I would appreciate if you could download the document into your website as appropriate.

Sincerely,

Prof. Anver Kuliev, MD, PhD